Author's response to reviews

Title: Access to Eye Health Services among Indigenous Australians: an Area Level Analysis

Authors:

Margaret Kelaher (mkelaher@unimelb.edu.au)
Angeline Ferdinand (afer@unimelb.edu.au)
Hugh Taylor (mkelaher@unimelb.edu.au)

Version: 6 Date: 15 May 2012

Author's response to reviews:

We have expanded on the ethical statement provided in the previous report. This study did not involve the collection of primary data from human subject. It involved the analysis of secondary data collected by statutory authorities. Access to these data and their ethical use is dictated by legislation and enforced by the data custodians. Details of the review process have been expanded.